Author’s response to reviews

Title: Is Daptomycin effective in the Central Nervous System infections? A case report.

Authors:

Fabrizio Taglietti (taglietti.f@gmail.com)
Floriana Campanile (campanile@unict.it)
Alessandro Capone (alessandro.capone@inmi.it)
Antonino Di Caro (antonino.dicaro@inmi.it)
Elisabetta Grilli (elisabetta.grilli@inmi.it)
Giulia Stazi (giulia.stazi@inmi.it)
Taschia Bertuccio (bertuccio@unict.it)
Nicola Petrosillo (nicola.petrosillo@inmi.it)
Stefania Stefani (stefanis@unict.it)

Version: 2 Date: 30 March 2012

Author’s response to reviews: see over
To Journal of Medical Case Reports

The Editor

RE: manuscript submission.

March 09, 2012

Dear Editor,

Attached please find our manuscript “Is Daptomycin effective in the Central Nervous System infections? A case report” to be submitted to for publication.

Our manuscript is an original contribution not previously published, and not under consideration for publication elsewhere. All the authors have seen and approved the submitted version of the paper.

Kind regards

Corresponding author:

Fabrizio Taglietti, M.D.

2nd Infectious Diseases Division,

National Institute for Infectious Diseases “L. Spallanzani”,

Via Portuense, 292-00149 Rome, Italy.

e-mail: fabrizio.taglietti@inmi.it

tel +390655170241

fax +390655170315